

# GOPEN ACCESS

**Citation:** Hill SC, de Souza R, Thézé J, Claro I, Aguiar RS, Abade L, et al. (2020) Genomic Surveillance of Yellow Fever Virus Epizootic in São Paulo, Brazil, 2016 – 2018. PLoS Pathog 16(8): e1008699. https://doi.org/10.1371/journal. ppat.1008699

**Editor:** Adam S. Lauring, University of Michigan, UNITED STATES

Received: March 10, 2020

Accepted: June 10, 2020

Published: August 7, 2020

**Copyright:** © 2020 Hill et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: YFV genome sequences are available on GenBank, under accession numbers MH030049-MH030090, MH193173- MH193175, and MT497520-MT497525.

**Funding:** NRF, ECS, OGP, SCH, NL and RdS were supported through a São Paulo Research Foundation (FAPESP) and Medical Research Council CADDE partnership award (MR/S0195/1 and FAPESP 18/14389-0) (http://caddecentre.org/). NRF and SCH were supported by a Wellcome Trust RESEARCH ARTICLE

# Genomic Surveillance of Yellow Fever Virus Epizootic in São Paulo, Brazil, 2016 – 2018

Sarah C. Hill<sup>1,2®</sup>\*, Renato de Souza<sup>3®</sup>, Julien Thézé<sup>1,4</sup>, Ingra Claro<sup>5</sup>, Renato S. Aguiar<sup>6</sup>, Leandro Abade<sup>1</sup>, Fabiana C. P. Santos<sup>3</sup>, Mariana S. Cunha<sup>3</sup>, Juliana S. Nogueira<sup>3</sup>, Flavia C. S. Salles<sup>5</sup>, Iray M. Rocco<sup>3</sup>, Adriana Y. Maeda<sup>3</sup>, Fernanda G. S. Vasami<sup>3</sup>, Louis du Plessis<sup>1</sup>, Paola P. Silveira<sup>6</sup>, Jaqueline G. de Jesus<sup>5</sup>, Joshua Quick<sup>7</sup>, Natália C. C. A. Fernandes<sup>3</sup>, Juliana M. Guerra<sup>3</sup>, Rodrigo A. Réssio<sup>3</sup>, Marta Giovanetti<sup>8</sup>, Luiz C. J. Alcantara<sup>8</sup>, Cinthya S. Cirqueira<sup>3</sup>, Josué Díaz-Delgado<sup>3</sup>, Fernando L. L. Macedo<sup>3</sup>, Maria do Carmo S. T. Timenetsky<sup>3</sup>, Regiane de Paula<sup>9</sup>, Roberta Spinola<sup>9</sup>, Juliana Telles de Deus<sup>10</sup>, Luís F. Mucci<sup>10</sup>, Rosa Maria Tubaki<sup>10</sup>, Regiane M. T. de Menezes<sup>10</sup>, Patrícia L. Ramos<sup>11</sup>, Andre L. de Abreu<sup>12</sup>, Laura N. Cruz<sup>3</sup>, Nick Loman<sup>7</sup>, Simon Dellicour<sup>13,14</sup>, Oliver G. Pybus<sup>1,2</sup>, Ester C. Sabino<sup>5</sup>\*, Nuno R. Faria<sup>1+</sup>\*

1 Department of Zoology, University of Oxford, Oxford, United Kingdom, 2 Department of Pathobiology and Population Sciences, Royal Veterinary College, Hawkshead, United Kingdom, 3 Instituto Adolfo Lutz, São Paulo, Brazil, 4 Université Clermont Auvergne, INRAE, VetAgro Sup, UMR EPIA, Saint-Genès-Champanelle, France, 5 Instituto de Medicina Tropical, Departamento de Moléstias Infecciosas e Parasitárias, Faculdade de Medicina e, Universidade de São Paulo, São Paulo, Brazil, 6 Laboratório de Virologia Molecular, Departamento de Genética, Instituto de Biologia, Rio de Janeiro, Brazil, 7 Institute of Microbiology and Infection, University of Birmingham, Birmingham, United Kingdom, 8 Laboratório de Flavivírus, Instituto Oswaldo Cruz, FIOCRUZ, Rio de Janeiro, Brazil, 9 Centro de Vigilância Epidemiológica "Prof. Alexandre Vranjac", São Paulo, Brazil, 10 Superintendência do Controle de Endemias, São Paulo, Brazil, 11 Fundação Parque Zoológico de São Paulo, São Paulo, Brazil, 12 Secretaria de Vigilância em Saúde, Ministério da Saúde (SVS/MS), Brasília-DF, Brazil, 13 Spatial Epidemiology Lab (SpELL), Université Libre de Bruxelles, CP160/12 50, Bruxelles, Belgium, 14 Department of Microbiology, Immunology and Transplantation, Rega Institute, KU Leuven, Leuven, Belgium

• These authors contributed equally to this work.

\* sarah.hill@zoo.ox.ac.uk (SCH); sabinoec@gmail.com (ECS); nuno.faria@zoo.ox.ac.uk (NRF)

# Abstract

São Paulo, a densely inhabited state in southeast Brazil that contains the fourth most populated city in the world, recently experienced its largest yellow fever virus (YFV) outbreak in decades. YFV does not normally circulate extensively in São Paulo, so most people were unvaccinated when the outbreak began. Surveillance in non-human primates (NHPs) is important for determining the magnitude and geographic extent of an epizootic, thereby helping to evaluate the risk of YFV spillover to humans. Data from infected NHPs can give more accurate insights into YFV spread than when using data from human cases alone. To contextualise human cases, identify epizootic foci and uncover the rate and direction of YFV spread in São Paulo, we generated and analysed virus genomic data and epizootic case data from NHPs in São Paulo. We report the occurrence of three spatiotemporally distinct phases of the outbreak in São Paulo prior to February 2018. We generated 51 new virus genomes from YFV positive cases identified in 23 different municipalities in São Paulo, mostly sampled from NHPs between October 2016 and January 2018. Although we observe substantial heterogeneity in lineage dispersal velocities between phylogenetic branches, continuous phylogeographic analyses of generated YFV genomes suggest that YFV and Royal Society Sir Henry Dale Fellowship awarded to NRF (grant 204311/Z/16/Z). The research was supported by an internal HEFCE GCRF grant 005073 and John Fell Research Fund Grant 005166, awarded to NRF. ECS was supported by grants CNPg #400354/2016-0 and FAPESP# 2016/01735-2. OGP, NRF and LdP were supported by the Oxford Martin School. SD is supported by the Fonds National de la Recherche Scientifique (FNRS, Belgium) and was previously funded by the Fonds Wetenschappelijk Onderzoek (FWO, Belgium). LCJA was supported by CNPg/ MCTI Decit/SCTIE/MoH (440685/2016-8) and CAPES (88887.130716/2016-00). MG was supported by Fundação de Amparo à Pesquisa do Estado do Rio de Janeiro – FAPERJ. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

lineages spread in São Paulo at a mean rate of approximately 1km per day during all phases of the outbreak. Viral lineages from the first epizootic phase in northern São Paulo subsequently dispersed towards the south of the state to cause the second and third epizootic phases there. This alters our understanding of how YFV was introduced into the densely populated south of São Paulo state. Our results shed light on the sylvatic transmission of YFV in highly fragmented forested regions in São Paulo state and highlight the importance of continued surveillance of zoonotic pathogens in sentinel species.

### Author summary

Since July 2016, the southeast region of Brazil has experienced the largest yellow fever virus (YFV) outbreak in decades. São Paulo is the most densely populated state in southeast Brazil. The outbreak has caused serious public health concern in the state, as YFV does not normally circulate widely there and most of the 21 million inhabitants were correspondingly unvaccinated against YFV when the outbreak began. In Brazil, YFV typically circulates among non-human primates (NHPs), and human cases represent isolated spill over events from this predominantly sylvatic cycle. Understanding the epidemiological dynamics and spread of YFV in NHPs is therefore critical for contextualising human cases, and guiding vaccination strategies that can better protect local human populations. Here, we aim to contextualise human cases, identify epizootic foci and uncover the rate and direction of YFV spread in São Paulo. We analyse the geographic and temporal distribution of observed cases of YFV in NHPs in São Paulo state, and identify three distinct phases of the epizootic. We generate virus genome sequences from 51 YFV-positive cases and perform phylogenetic and phylogeographic analyses aimed at understanding the spatial spread of YFV in São Paulo state. Analyses of these data indicate that YFV spread from the north of São Paulo state into more densely populated southern regions. Although we observe substantial heterogeneity in the rate at which different sampled YFV lineages spread, the typical rate of spread was low with a mean rate of  $\sim 1$  km per day. This is consistent with a scenario in which the majority of transmission events occurred between NHPs primates and sylvatic vectors across forested patches.

# Introduction

Yellow fever (YF) is an acute haemorrhagic disease caused by yellow fever virus (YFV), a single-stranded positive-sense RNA virus from the *Flavivirus* genus. Yellow fever is endemic to the American and African tropics, where nearly 400 million people are estimated to be at risk of infection [1]. Clinical manifestations of YF in humans range from inapparent or mild disease in up to 80% of infected cases, to severe hepatitis and haemorrhagic disease. The fatality rate among patients who develop visceral disease can range from 20% to 60% [2]. YF is preventable in humans by administration of a single dose of an extremely effective vaccine that provides life-long protection against the disease.

YFV is primarily transmitted in the Americas in a sylvatic cycle between non-human primates (NHPs) and tree-dwelling mosquitoes (primarily *Haemagogus janthinomys*, *Haemagogus leucocelaenus* and more rarely mosquitoes of the *Sabethes* genus) [3–5]. The urban cycle of YFV, in which YFV is transmitted between humans and *Aedes aegypti* mosquitoes, has not been observed in the Americas since the 1940s [2,6]. In Brazil, outbreaks typically initiate amongst NHPs in neotropical forests and cause subsequent human epidemics that mostly affect unvaccinated male adults in rural areas. YFV infection tends to be detected earlier in NHPs before occasional human spill over infections are observed, and infection location is typically more precisely ascertained for NHPs than for humans. Surveillance of primate epizootics is therefore critical to alert us to the possibility of future human infections [7], and to provide the clearest possible indicator of spatiotemporal disease spread.

From July 2016, the southeast region of Brazil suffered the largest YFV outbreak observed in the Americas in decades. The epizootic/epidemic was caused by a rapidly-spreading lineage of the South American 1 (SA1) genotype that is thought to have originated in the Amazon basin [8]. Between December 2016 and June 2019, the Brazilian Ministry of Health confirmed at least 2251 human YFV cases and 1567 epizootic events in NHPs across Brazil [9]. São Paulo state is a densely populated state of southeast Brazil that contains one of the world's largest urban conurbations. Available reports from São Paulo suggest that the state health authorities confirmed around 875 cases of YFV in NHPs between July 2016 and 18<sup>th</sup> November 2019 and 624 cases of YFV in humans between January 2017 and 18<sup>th</sup> November 2019 [10–12]. YFV genomes from 36 human cases sampled during the outbreak have recently been reported from São Paulo [13]. However, despite the magnitude and expansion of the epizootic in São Paulo, remarkably little is known about the origins, spread or genetic diversity of YFV in NHP populations in the region.

Analytical insights into the spread of YFV among NHP populations can guide public health efforts more effectively than those based on human cases alone. This is because routes and rate of YFV spread estimated solely from human cases can be biased by substantial human travel away from the location of infection, or obscured by the lack of human cases in well-vaccinated regions where epizootics are nevertheless ongoing. Epizootic foci can be used to determine which human populations may be most urgently in need of vaccination. Understanding how YFV is introduced into new areas, particularly highly urbanised areas such as São Paulo city, is important for designing strategies that can interrupt these introductions. Estimates of the rate of YFV spread can help public health organisations ensure healthcare preparedness is achieved before the anticipated onset of local human cases. To address these questions, we report case data from NHP populations in São Paulo from January 2016—February 2018. We sequence YFV from NHPs in São Paulo using a previously-developed portable sequencing approach. We analyse case data and conducted virus phylogeographic analyses to characterize the spread of YFV in São Paulo during the first waves of the outbreak.

#### Methods

#### **Ethics statement**

Research on human cases was supported by the Brazilian Ministry of Health as part of arboviral genomic surveillance efforts within the terms of Resolution 510/2016 of CONEP (Comissão Nacional de Ética em Pesquisa, Ministério da Saúde; National Ethical Committee for Research, Ministry of Health). Only naturally deceased NHPs were sampled for YFV surveillance. The surveillance protocol for dead NHPs was approved by the Ethics Committee for the use of Animals in Research, Instituto Adolfo Lutz, under the numbers 0135D/2012 and 020G/2014.

#### Sample collection and mapping of non-human primate cases

Notifications of dead NHPs were made to the São Paulo State Epidemiological Health Department through the Information System for Notifiable Diseases (SINAN). Quantum GIS [14] was used to create a choropleth map depicting the geographic distribution and number of confirmed cases across different municipalities in São Paulo. The number of reported NHPs varied over time in both northern and southern parts of the state (S1 Fig).

#### Identification of YFV positive cases

Reported NHP carcasses were tested for YFV. Not all reported carcasses were tested for YFV and therefore the number of positive animals reported here likely underestimates YFV incidence in NHPs in São Paulo. The proportion of all cases in our data changed over time, such that proportionally fewer reported NHPs were tested in late 2017 and early 2018 than at an early stage of outbreak emergence (S1 Fig). Despite this, the proportion of all carcasses that were tested following reporting was relatively consistent across each municipality (S2 Fig).

YFV was confirmed by a positive result in at least one of the four methods described below; immunohistochemistry, immunofluorescence, reverse transcription quantitative PCR (RT-qPCR), or epidemiological linkage.

#### Histopathology and immunohistochemistry for NHP cases

All samples were subjected to histopathology and immunohistochemistry examination. Samples of brain, heart, lung, liver, spleen and kidney from NHPs were fixed in formaldehyde and embedded in paraffin. Histological sections of these tissues were stained with haematoxylin and eosin and examined on a microscope. Indirect immunohistochemistry was used to detect the presence of the YFV antigen. Sections of liver (0.3uM) were placed on slides coated with silane and were treated with an in-house polyclonal anti-YFV antibody that was produced in mice and used at 1/30,000 dilution. The slides were treated with anti-mouse secondary antibodies, linked to either horseradish peroxidase (Reveal HRP Spring polymer System, Spring) or to alkaline phosphate (Link MACH4 universal AP Polymer and Polymer MACH4 universal AP, Biocare). Chromogenic detection of the presence of YFV was subsequently conducted using the substrates 3,3'-diaminobenzidine or Fast Red (Warp Red, Biocare), respectively.

#### Indirect immunofluorescence

Samples of NHP blood or serum, and tissue material suspensions obtained from autopsies, were tested using a standardized indirect immunofluorescence technique [15]. An in-house polyclonal anti-flavivirus antibody (anti-DENV 1–4) and an anti-mouse IgG-FITC antibody (Sigma) were used. Positive samples were typed by indirect immunofluorescence with monoclonal antibodies for YFV (Biomanguinhos, Rio de Janeiro).

#### **RT-qPCR**

Total RNA was extracted from tissue and serum samples using two commercial kits according to the manufacturer's instructions: QIAamp RNA Blood Mini Kit for tissues and QIAamp Viral RNA Mini Kit for serum (Qiagen Inc., Germany). Viral RNA was detected using two previously published RT-qPCR techniques [16,17].

#### Epidemiological linkage

Where multiple NHPs from the same species were found dead at the same time and location, the number of carcasses was recorded but typically only one or a few animals were tested. In these instances, untested animals were on rare occasions considered as confirmed cases on the basis of their spatiotemporal association with tested, confirmed cases.

#### Sample collection of human cases

We randomly selected 5 samples from confirmed human cases collected in municipalities of São Paulo state (S1 Table) that had been tested using the RT-qPCR methods, above. These samples were collected in several municipalities of São Paulo state and sent for molecular diagnostics at Instituto Adolfo Lutz.

#### MinION genome sequencing

A selection of positive samples (S3 Fig and S1 Table) was sequenced using a rapid wholegenome sequencing protocol that has been previously validated and successfully applied in Brazil [8,18]. NHP samples were selected to have RT-qPCR cycle threshold (CT) values <25 to facilitate genomic amplification and sequencing. Because few samples were available, human samples were randomly selected from positive cases with CT <37. In brief, cDNA was produced from viral RNA using random hexamers and the Protoscript II First Strand cDNA synthesis kit (NEB). The genome was amplified using a multiplex PCR scheme designed to produce overlapping 500bp amplicons across the whole coding region of the recent South American genotype I outbreak clade [8,18]. PCR products were quantified, barcoded using the Oxford Nanopore Technologies (ONT) Native Barcoding Kit (NBD103, or NBD103 and NBD114, depending on sequencing date), and pooled in an equimolar fashion. Sequencing libraries consisting of 10–24 samples per library were constructed using a Ligation Sequencing kit (ONT, SQK-LSK108 or SQK-LSK109, depending on sequencing date). Sequencing was performed on a ONT FLO-MIN106 flow cell for up to 48h as described previously [8].

#### Generation of consensus sequences

Previously published approaches were used to produce YFV consensus sequences [18,19]. In brief, raw files were basecalled using Albacore or Guppy version 3.0.3 (Oxford Nanopore Technologies). Reads were demultiplexed and trimmed of adaptor and barcode sequences using Porechop (version 0.2.3, https://github.com.rrwick/Porechop). Trimmed reads for each barcoded sample were mapped to a reference genome (accession number JF912190) using *bwa* [20], and primer binding sites were trimmed from reads. Single nucleotide variants to the reference genome were detected using nanopolish variant calling (version 0.11.1) [21]. Consensus sequences were generated using identified variants. Sites with <20X coverage were replaced with ambiguity code N. Sequencing statistics can be found in <u>S2 Table</u>. Accession numbers of newly generated sequences can be found in <u>S1 Table</u>.

#### **Phylogenetic analyses**

The genome sequences generated here (n = 51) were combined with all previously published SA1 genotype YFV genomes available to 31 December 2019 (n = 173). This dataset was curated to remove all sequences shorter than 70% of the whole coding sequence length, those with missing metadata, and those with abnormal genomic sequences typical of sequencing or assembly errors. The full dataset analysed here therefore contains 224 YFV sequences, as shown in S4 Fig. Sequences were aligned using MAFFT v.7 [22]. The best-fit substitution model was selected using ModelTest-NG [23]. Maximum likelihood (ML) phylogenetic trees were estimated using RAxML v.8 [24] under a GTR + I +  $\Gamma_4$  nucleotide substitution model. Statistical support for phylogenetic nodes was estimated using a ML bootstrap approach with 100 replicates.

#### Phylogeographic analyses

To investigate the spread of YFV in São Paulo, we analysed the São Paulo outbreak clade in more detail. We focus on two datasets; the 'full' São Paulo outbreak clade, represented by 99 sequences sampled from 2016–2018, and a 'geographically restricted' dataset in which 4 sequences are removed that appear to be clear spatial exports from the main São Paulo epizo-otic to distant locations in Goiás, Espírito Santo and Minas Gerais, with little phylogenetic evidence of extensive local transmission. These spatial outliers were removed so that rare, long-distance movements with little or no observed onwards transmission did not impact estimates of branch dispersal rate or maximal wavefront distance.

Geo-referenced and time-stamped sequences were analysed in BEAST v.1.10.4 [25] using the BEAGLE library to enhance computational speed [26]. The uncorrelated lognormally distributed relaxed clock model [27], a skygrid coalescent tree prior [28] and a continuous phylogeographic model that uses a relaxed random walk (RRW) to model the spatial diffusion of lineages were used. Contrary to a standard Brownian diffusion model that is based on a strong assumption of homogeneous dispersal velocity in all phylogenetic branches, the RRW model is a flexible model that specifically allows inference of heterogeneous, branch-specific dispersal velocities. Dispersal velocity variation among lineages was here represented using a Cauchy distribution [29,30]. The default settings were used for the diffusion rate. Uniform priors were used to allow date uncertainty for those sequences where exact sampling day was unknown, and a jitter added to tips with non-unique co-ordinates (window size of 0.2 decimal degrees).

Virus diffusion through time and space was summarised using 1000 phylogenies sampled at regular intervals from the posterior distribution (after exclusion of 10% burn-in). The R package "seraphim" [31] was used to extract the spatiotemporal information contained in these phylogenies, whose branches can be considered as movement vectors (each having start and end spatial coordinates, and start and end dates, in decimal units). The package was also used to estimate statistics of spatial dissemination, such as mean branch dispersal velocity, change in the maximal wavefront distance from epizootic origin, and an estimate of subsequent wavefront dispersal velocity [31]. The maximum wavefront distance is the distance between the epidemic origin (here approximated as the inferred tree root location) and the furthest location at which virus transmission has ever occurred. Wavefront velocity is the rate at which the maximum wavefront distance increases over time. In contrast, branch dispersal velocity captures the rate of movement of any lineage, regardless of whether or not it expands the epidemic wavefront. The spatiotemporal diffusion of YFV was visualised using the "spreadGraphic2" R function [31].

#### Results

Liver, brain or blood samples from 591 NHPs were considered positive for YFV between epidemiological week 29 of 2016 to week 4 of 2018. Samples were confirmed as YFV positive cases using immunohistochemistry (n = 268), RT-qPCR (n = 113), both immunohistochemistry and RT-qPCR (n = 198) or in rare instances by epidemiological linkage to another positive NHP (n = 12). Most confirmed YFV-positive NHPs were animals of the *Alouatta* genus (88%; 403 of 459 cases for which genus information was available), followed by *Callithrix* (8%; 35/459), *Callicebus* (2%; 9/459), *Cebus* (2%; 9/459) and *Sapajus* (0.7%; 3/459).

Our results reveal three main epizootic phases, including a phase of low reported NHP case counts in northern São Paulo state, followed by two phases where higher numbers of NHP cases were reported almost exclusively in the south the of state. Phase 1 runs from July 2016 to Jan 2017 (6%, 33/591 cases). Phase 2 runs from February to July 2017 (20%, 119/591 cases) and phase 3 from July 2017 to at least February 2018 (74%, 439/591 cases) (Fig 1A). These



**Fig 1. Distribution of weekly YFV cases. A)**. NHP YFV positive confirmed cases per week. **B**). Choropleth map of São Paulo municipalities reporting positive NHP YFV cases during each outbreak phase. Shapefiles used to produce these maps are freely available from the Brazilian Institute of Geography and Statistics (IBGE) [42].

https://doi.org/10.1371/journal.ppat.1008699.g001

transition dates were chosen to capture the spatial shift between phase 1 and phase 2 (**Fig 1B**), and to occur centrally within the low case count period that occurs between phase 2 and 3 (**Fig 1A**), but the exact dates adopted here are somewhat arbitrary. The peaks of epizootic phases 2 and 3 were mid-April 2017 and late November 2017, respectively. Proportionally fewer NHP carcasses were tested in phase 3 than in phase 1 or 2, and in January and February 2018 than in many earlier months (**S1 Fig**). It is therefore possible that phase 3 may have been of longer duration and proportionally greater magnitude than we report here.

NHP cases were geographically widespread across 57 municipalities of São Paulo to February 2018, with most reported cases concentrated around the southeast region of the state (Fig 2). The greatest number of confirmed cases were observed in the municipalities of Mairiporã (n = 88), Jundiaí (n = 68), São Paulo (n = 62), Bragança Paulista (n = 48), and Atibaia (n = 36). There is a clear distinction between the geographic location of cases in epizootic phase 1, and those in phases 2 and 3 (Fig 1B). Specifically, almost all cases that form part of phase 1 occur in a geographically distinct cluster of municipalities in the north of São Paulo state (Fig 1B; Fig 2). On the other hand, nearly all cases of YFV detected after the first phase occurred in

municipalities in the south of São Paulo state and within 200km of São Paulo city. (Fig 1B and Fig 2).

To investigate the source and transmission of YFV during all three epizootic waves, and to characterise the genetic diversity of the virus circulating in NHP populations across São Paulo state, we generated whole YFV genomes for 46 RT-qPCR positive NHP samples (median Ct-values of 14, range 9–25) from 20 municipalities using a previously described MinION sequencing protocol [8,18]. We also sequenced five whole genomes from human samples collected in four municipalities of São Paulo state (median Ct-values of 34, range 32–37). In total, this represents genomes from 51 samples collected across 23 municipalities between October 2016 and January 2018 (S3 Fig).

We first used maximum likelihood methods to estimate phylogenetic relationships amongst all 224 sequences available from SA1 in South America. Our genetic analyses indicate a strong phylogenetic spatial structure of the ongoing epizootic in São Paulo state, with 43 of 46 sequences from NHPs in São Paulo clustering in a single clade (bootstrap score = 98%), previously named as lineage YFV<sub>MG/SP</sub> [32]. This clade contains sequences sampled in São Paulo state, and an additional 8 sequences sampled in Minas Gerais that were all from within 100km of São Paulo state (Fig 3, see Fig 2 for locations of samples from Minas Gerais and S4 Fig for the full phylogeny containing all sequences).

Sequences from 6 NHPs sampled during the first epizootic phase (October to December 2016) form two separate phylogenetic clades in the maximum likelihood tree, one of which appears basal to the 2016–2019 outbreak clade in southeast Brazil. Whilst bootstrap support for this structure is relatively poor (bootstrap score = 67%), its presence raises the possibility that two distinct YFV lineages may have co-circulated in northern São Paulo state during the first low-intensity epizootic phase in late 2016.



**Fig 2. Choropleth map of the distribution of confirmed NHP cases per municipality in São Paulo state between July 2016 and February 2018.** Municipalities are shaded based on reporting and testing of NHPs. Circles depict locations of those cases from humans that were sequenced as part of this study, and red and orange triangles depict locations in São Paulo state from which NHP genome sequences were sequenced as part of this study. Eight samples from 7 municipalities in MG that form a clade with our sequences from São Paulo and that were sequenced by previous studies [8,43] are shown in purple triangles. Samples SA131 and Y37-244 are indicated. Shapefiles used to produce this maps are freely available from the Brazilian Institute of Geography and Statistics (IBGE) [42].

https://doi.org/10.1371/journal.ppat.1008699.g002



**Fig 3. Maximum likelihood phylogenetic tree of YFV in Brazil.** Nodes with >50% bootstrap support are coloured dark blue, and bootstrap scores are provided for highly supported nodes (> = 75%). Sequences generated in this study have black tip labels (n = 51). One clade (previously denoted lineage YFV<sub>MG/ES/RJ</sub> [32]) is collapsed for clarity. This clade contains three sequences that were sampled from humans in São Paulo, but for all three cases patient travel histories indicate that YFV was acquired elsewhere. Samples SA131 and Y37-244 are indicated with an asterisk and discussed in the Discussion.

https://doi.org/10.1371/journal.ppat.1008699.g003

We used a continuous diffusion model to investigate the spatiotemporal spread of YFV. Our analyses were unable to accurately determine the route via which YFV was introduced to São Paulo state. Although the root of the MCC tree is placed in north-western MG (red dot, **Fig 4**), the spatial uncertainty in this root location is very high. In addition, posterior support for phylogenetic grouping of sequences from Goiás state with those from northern São Paulo is very weak (0.12), and as such this reconstructed lineage movement should be interpreted with caution. Whilst we were unable to identify the route of introduction to northern São Paulo, our analyses are better able to resolve subsequent viral dispersal trajectories from the north to the south of the state. Specifically, YFV appears to have disseminated from northern São Paulo into geographically neighbouring areas of western MG and into the south of the state.

Both those analyses including and discarding spatially-outlying sequences (see Fig 4 for locations of the discarded sequences) suggest that the mean branch velocity was ~1 km/day (geographically restricted dataset: 0.83 km/day, 95% highest posterior density (HPD), 0.50–1.53, full dataset: 0.95 km/day, 95% HPD 0.53–2.42). The mean velocity did not vary significantly between the three epizootic phases observed in São Paulo (S3 Table). There is, however, heterogeneity in the dispersal velocity of different branches, such that most phylogenetic branches have lineage dispersal velocities that are somewhat lower than the mean, and a few are associated with rapid dispersal (Fig 5).

Considering all sequences, the wavefront velocity varies across time (Fig 6). However, rare distant outlier sequences can heavily inform the maximum wavefront distance, even if associated expansive virus lineages do no contribute substantially to the overall epizootic and only represent isolated cases in sampled outlier locations. Considering sequences in the spatially



**Fig 4. Reconstructed spatiotemporal diffusion of YFV in São Paulo.** Phylogenetic branches are mapped in space according to the location of phylogenetic nodes and tips (circles). Shaded regions coloured according to time show 80% highest posterior density contours calculated using bivariate kernel density estimation of the location of all nodes observed within defined 6-month time intervals in a subset of 1000 trees. Illustrated nodes are from the MCC tree and are coloured according to time. A) Spatiotemporal diffusion of taxa based on analysis of the full dataset (n = 99 sequences). B) Expansion of region highlighted in **A** without the long-distance outlier sequences. Shapefiles used to produce this map are available in the public domain from Natural Earth (https://www.naturalearthdata.com/).

https://doi.org/10.1371/journal.ppat.1008699.g004



**Fig 5.** Density of branch dispersal velocity in the entire (**panel A**) and restricted (**panel B**) datasets (**see Fig 4**). Branch dispersal velocities are natural log transformed. Each individual trace represents branch dispersal velocity density of one tree from the generated posterior distribution of phylogenetic trees.

https://doi.org/10.1371/journal.ppat.1008699.g005

restricted area only (see Fig 4 for locations), the posterior intervals on the wavefront velocity are relatively wide and we therefore cannot exclude that the wavefront velocity of the majority of the sampled epizootic is homogeneous across time (Fig 6). In this dataset, the epidemic wavefront moved on average only 0.4 km/day (95% HPD, 0.30–0.53) during January 2015 to January 2017 (Fig 6).

#### Discussion

In this study we analyse data on the epizootic intensity and spatiotemporal distribution of YFV cases in NHPs in São Paulo state, Brazil, during 2016–2018. Our results demonstrate the existence of three distinct phases of the epizootic to early 2018. Specifically, we observe an



**Fig 6.** Spatial wavefront distance from epidemic origin over time for all studied sequences (grey), and those in the restricted area shown in **Fig 4B** (green), as estimated with the "seraphim" package. The identified epizootic phases are indicated by bars in the upper right. Ribbons show 95% HPDs.

#### https://doi.org/10.1371/journal.ppat.1008699.g006

initial phase in which a small number of cases were identified in the northern region of São Paulo state during late 2016. This period was followed by two larger epizootic phases; the first from February 2017 to July 2017, and the second from July 2017 to at least February 2018 (the date of the most recent data available for this study). During the second and third phases, almost all cases were observed in the south of São Paulo state. We generated 51 novel YFV genomes from samples collected in 23 different municipalities in São Paulo state. The majority of sequenced viruses were sampled from NHPs of the genus *Alouatta* from late 2016 to early 2018, and represent the first and third epizootic phases identified here (S3 Fig, Fig 1). Phylogeographic analyses of these data indicate that after being introduced to the north of São Paulo state, YFV lineages spread southwards and initiated epizootics in the south of the state. Lineages spread at a mean branch velocity of approximately 1km per day (Fig 4 and Fig 5), though there is substantial heterogeneity in branch dispersal velocity (Fig 5).

Previous analyses have hypothesised that the epizootic in the densely populated south of São Paulo originated in Minas Gerais [13], but those analyses were performed without any data originating from the north of São Paulo state. Here, we revise this hypothesis and suggest that the epizootic in the densely populated south of São Paulo (epizootic phase 2 and 3) may instead have originated in the north of the state. However, we note that few YFV-positive carcasses were identified in the centre-east of São Paulo (**Fig 2**) and genomes sampled from NHP in the north and south of São Paulo are connected by a single branch in the phylogenetic tree, with no genomes sampled in between. As such, it is impossible to identify the exact route or mode of YFV spread through the state. Several possibilities might explain the movement of YFV between northern and southern São Paulo: (i), YFV-positive NHPs were present but were not identified in a corridor in the eastern São Paulo region due to limited sampling (**Fig 2**), (ii), YFV moved via an as-yet-unsampled 'stepping-stone' NHP populations in neighbouring MG, or (iii), YFV was directly transmitted between the two locations through human activities, without the infection of intermediate NHP populations. The northern regions of São Paulo that reported YFV in 2016–2017 are similar to those affected by YFV in 2008 and 2000 [33,34],

and it is therefore plausible that YFV will emerge again there in future. Understanding how and why the outbreak spread southwards more extensively from northern São Paulo during the 2016–2019 outbreak than in previous outbreaks could help inform future surveillance and control efforts.

Determining how YFV is introduced into highly urbanised areas, such as São Paulo city, is important for designing strategies that can effectively interrupt such introductions. One of our sequenced samples, SA131 (accession number MH193175), was collected from an Alouatta sp. individual at the Parque Estadual das Fontes do Ipiranga (PEFI) (Fig 2 and Fig 3). This park forms part of São Paulo city's zoo, and represents a 53 km<sup>2</sup> fragmented island of Atlantic forest within the urbanised city metropolis. Surprisingly, phylogenetic analyses show that the YFV genome recovered from sample SA131 clusters together with isolate Y37-244 (accession number MH030085) (bootstrap score for ML tree = 68%, 95% posterior support for MCC tree = 1), collected in Piracaia, 75 km from the park, and not with any of the isolates that were detected in the outskirts of São Paulo (Fig 2 and Fig 3). This result could be explained by: (i) incomplete sampling of a wave of continuous transmission among NHPs that live between the two locations; (ii) human-mediated transport of YFV infected NHPs [35], (iii) human-mediated transport of mosquitos, or (iv) introduction of the virus by an asymptomatic human visitor, who carried the virus from Piracaia to PEFI. Scenario (i) seems unlikely since PEFI is an isolated forested fragment with limited connectivity to other forested areas; at this stage, scenarios (ii), (iii) and (iv) all remain possible.

Using YFV genetic data, we estimate that virus lineages dispersed at a mean rate of ~1 km per day during the outbreak in São Paulo (95% HPD, 0.50–1.53). We did not observe substantial differences in the mean lineage dispersal between epizootic phases (S3 Table), which might have indicated different impact of drivers in different epizootic phases or corresponding affected regions. However, our analyses demonstrated substantial heterogeneity between different phylogenetic branches (Fig 5). Most phylogenetic branches show low lineage velocity (<3 km/day) that are consistent with YFV spread by NHPs and mosquitos between contiguous forested patches. Far more rarely, we observe higher velocity lineage movements. Such movement of infected humans or mosquitos. We also cannot rule out that terminal phylogenetic branches with unusually high lineage dispersal velocity may have been caused by accidental attribution of incorrect metadata to samples, and efforts should be made to perform confirmatory sequencing of outlier samples and sequence YFV from other samples collected in their vicinity. Greater genomic sampling density is required to better investigate the drivers of such rapid YFV lineage movements.

Our mean estimate of the rate of YFV lineage movement is slightly slower than that those previously estimated for the states of Minas Gerais, Espírito Santo and Rio de Janeiro, and for the southeast region as a whole (~4km/day) [8,13]. Phylogenetic estimations of rate of spread often depend on sampling range and tend to be biased when rates are estimated for large geographic areas [36]. However, the relatively confined area of YFV circulation suggests that relaxed random walk models used here are a good approximation for viral diffusion in the southeast region of Brazil. Additional analyses of larger datasets are necessary to test whether spatial heterogeneity in NHP density, habitat fragmentation, or impact of human activities, may be responsible for the differences observed in the rate of YFV spread among different areas of southeast Brazil.

The NHP case data presented here (Fig 1 and Fig 2) suggest that the epizootic was larger in the southern parts of São Paulo than in the north. We hypothesise that this could be related to different NHP or vector species density or distribution in each area. Specifically, the southern part of São Paulo has large, connected forested areas that support a high diversity and density of NHPs [37,38]. In contrast, the north of São Paulo is covered by savannah and semi-deciduous forest [38], which may offer more restricted habitat availability for YFV vectors and amplifying NHP species.

Despite this, the magnitude of the YFV epizootic presented in Fig 1 and Fig 2 is based on epidemiological surveillance alone, and we caution that any bias in reporting or testing of NHP carcasses would also bias these results. Multiple municipalities reported small numbers of NHP carcasses that could not be tested, and/or tested relatively few carcasses in total. It is therefore possible that municipalities that did not report positive cases (Fig 1) nevertheless had local, undetected epizootics. YFV surveillance based on reporting of dead animals may have led to the appearance of a larger YFV epizootic in the south of São Paulo São Paulo, because animals from the Alouatta genus that are more likely to die of YFV infection than other genera [39] are more common in the south of São Paulo [37] than the north. The south of São Paulo is more densely human-populated than the north, so we also cannot exclude that dead NHPs would have been noticed and reported more frequently there. This could lead to the appearance that the outbreak was larger in the south. Reporting of NHP carcasses varies over time (S1 Fig). It is impossible to determine to what extent time-vary reporting of NHP carcasses was caused by consistent reporting of time-varying NHP deaths, or by changing frequency at which NHP deaths were reported by human observers. However, our observation that the epizootic in northern São Paulo was largely extinguished after phase 1 is unlikely to have been severely affected by such surveillance biases, as northern locations continue to test NHP carcasses after this time without detecting positive samples (Fig 1, S1 Fig). This finding is also consistent with the spatial distribution of YFV cases in humans [40]. Greater efforts to sample mosquito vectors or live NHP during epizootic periods would be important to determine whether reliance on reporting of NHP carcasses biases our understanding of YFV spatial distribution.

To facilitate sequencing, we preferentially studied samples showing low CTs during RTqPCR. Our selected samples were therefore biased towards those obtained from animals with higher viremia, and hence likely towards animals with greater clinical disease and from the *Alouatta* genus. Further, all phylogeographic inference is conditioned upon sampling, such that we can only reconstruct lineage movement amongst those lineages that we sample. The potential sampling biases that we describe above may have resulted in major or minor differences between the actual viral spread and the inferred dispersal history of lineages [41]. Analysis of larger YFV genomic datasets incorporating more sequences from non-*Alouatta* genera (including humans with insubstantial travel history), animals with lower viremia, and animals from genomically unsampled locations is critical to determine whether this could have affected our phylogeographic reconstructions of YFV spread in southeast Brazil.

Our study sheds light on the spatial and temporal dynamics of YF in NHP hosts across São Paulo state. A better understanding of the vectors and host species involved in the persistence of the virus, denser genomic sampling of available positive samples, and targeted investigation into phylogenetic branches associated with long-distance or rapid movements, will be important to understand the drivers of YFV transmission and anticipate future outbreaks.

#### Supporting information

**S1 Fig.** Reporting and testing of NHPs over time in northern (**panel A**) and southern (**panel B**) regions of São Paulo state. Here, municipalities where centroids fall north of latitude -21.54 are considered 'northern', and others are considered 'southern' (latitude indicated on **panel C**, along with locations of municipalities with positive cases in grey). This latitude was chosen to discriminate between locations most affected during phase 1, and those most affected during

phase 2 and 3, and is therefore not central in São Paulo. Colours indicate results of testing. Darker shades of each colour in each bar represent results in testing in any municipality that detected positive NHPs during any month, and paler shades represent cases from those municipalities that never detected positive cases during the displayed period. Shapefiles used to produce this map are freely available from the Brazilian Institute of Geography and Statistics (IBGE) [42]. (TIF)

S2 Fig. Number of carcasses tested and untested for YFV per municipality.

(TIF)

**S3** Fig. Date of sample collection and number of genomes generated from each host genus. (TIF)

S4 Fig. Maximum likelihood phylogeny of newly generated genomes sequences and all publicly available SA1 YFV sequences from Brazil (n = 224). (TIF)

**S1 Table. Details of the YFV genomes generated in this study.** All sequences are from São Paulo. (DOCX)

**S2 Table. Non-human primate YFV genome sequences from São Paulo, by host genus.** Detailed information for each sequenced isolate can be found in <u>S1 Table</u>. (DOCX)

**S3** Table. Mean branch dispersal velocity estimates. (DOCX)

## Acknowledgments

We thank the IAL and the USPTM staff, in particular the team from Núcleo de Doenças de Transmissão Vetorial and from the Centro de Patologia.

## **Author Contributions**

Conceptualization: Sarah C. Hill, Renato de Souza, Ester C. Sabino, Nuno R. Faria.

Data curation: Sarah C. Hill, Renato de Souza, Julien Thézé.

Formal analysis: Sarah C. Hill, Louis du Plessis, Simon Dellicour.

Funding acquisition: Nuno R. Faria.

- Investigation: Sarah C. Hill, Renato de Souza, Julien Thézé, Ingra Claro, Renato S. Aguiar, Leandro Abade, Fabiana C. P. Santos, Mariana S. Cunha, Juliana S. Nogueira, Flavia C. S. Salles, Iray M. Rocco, Adriana Y. Maeda, Fernanda G. S. Vasami, Paola P. Silveira, Jaqueline G. de Jesus, Natália C. C. A. Fernandes, Juliana M. Guerra, Rodrigo A. Réssio, Marta Giovanetti, Cinthya S. Cirqueira, Josué Díaz-Delgado, Fernando L. L. Macedo, Roberta Spinola, Juliana Telles de Deus, Luís F. Mucci, Rosa Maria Tubaki, Regiane M. T. de Menezes, Patrícia L. Ramos, Andre L. de Abreu, Laura N. Cruz.
- Methodology: Sarah C. Hill, Renato de Souza, Fabiana C. P. Santos, Mariana S. Cunha, Joshua Quick, Natália C. C. A. Fernandes, Juliana M. Guerra.

Project administration: Sarah C. Hill, Ingra Claro, Ester C. Sabino, Nuno R. Faria.

**Resources:** Renato de Souza, Maria do Carmo S. T. Timenetsky, Regiane de Paula, Roberta Spinola, Patrícia L. Ramos, Ester C. Sabino, Nuno R. Faria.

Software: Julien Thézé, Nick Loman, Simon Dellicour.

Supervision: Luiz C. J. Alcantara, Oliver G. Pybus, Ester C. Sabino, Nuno R. Faria.

Validation: Sarah C. Hill.

Visualization: Sarah C. Hill, Leandro Abade, Simon Dellicour, Nuno R. Faria.

Writing - original draft: Sarah C. Hill, Renato de Souza, Nuno R. Faria.

Writing – review & editing: Sarah C. Hill, Renato de Souza, Louis du Plessis, Simon Dellicour, Oliver G. Pybus, Nuno R. Faria.

#### References

- Shearer FM, Moyes CL, Pigott DM, Brady OJ, Marinho F, Deshpande A, et al. Global yellow fever vaccination coverage from 1970 to 2016: an adjusted retrospective analysis. Lancet Infect Dis. 2017; 17: 1209–1217. https://doi.org/10.1016/S1473-3099(17)30419-X PMID: 28822780
- Monath TP, Vasconcelos PFC. Yellow fever. Journal of Clinical Virology. 2015; 64: 160–173. <u>https://doi.org/10.1016/j.jcv.2014.08.030 PMID: 25453327</u>
- Vasconcelos PFC, Sperb AF, Monteiro HAO, Torres MAN, Sousa MRS, Vasconcelos HB, et al. Isolations of yellow fever virus from Haemagogus leucocelaenus in Rio Grande do Sul State, Brazil. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2003; 97: 60–62. <u>https://doi.org/10.1016/s0035-9203(03)90023-x</u> PMID: 12892055
- Vasconcelos PF da C. Yellow fever in Brazil: thoughts and hypotheses on the emergence in previously free areas. Revista de Saúde Pública. 2010; 44: 1144–1149. https://doi.org/10.1590/s0034-89102010005000046 PMID: 21109907
- Abreu FVS de, Ribeiro IP, Ferreira-de-Brito A, Santos AAC dos, Miranda RM de, Bonelly I de S, et al. Haemagogus leucocelaenus and Haemagogus janthinomys are the primary vectors in the major yellow fever outbreak in Brazil, 2016–2018. Emerging Microbes & Infections. 2019; 8: 218–231. https://doi.org/ 10.1080/22221751.2019.1568180 PMID: 30866775
- Souza RP de, Petrella S, Coimbra TLM, Maeda AY, Rocco IM, Bisordi I, et al. Isolation of yellow fever virus (YFV) from naturally infected Haemagogus (Conopostegus) leucocelaenus (diptera, cukicudae) in São Paulo State, Brazil, 2009. Rev Inst Med Trop Sao Paulo. 2011; 53: 133–139. <u>https://doi.org/10. 1590/s0036-46652011000300004</u> PMID: 21755235
- Vasconcelos PF da C. [Yellow Fever]. Rev Soc Bras Med Trop. 2003; 36: 275–293. <u>https://doi.org/10.1590/s0037-86822003000200012 PMID: 12806465</u>
- Faria NR, Kraemer MUG, Hill SC, Goes de Jesus J, Aguiar RS, Iani FCM, et al. Genomic and epidemiological monitoring of yellow fever virus transmission potential. Science. 2018; 361: 894–899. <u>https://doi.org/10.1126/science.aat7115</u> PMID: 30139911
- Silva NIO, Sacchetto L, de Rezende IM, Trindade G de S, LaBeaud AD, de Thoisy B, et al. Recent sylvatic yellow fever virus transmission in Brazil: the news from an old disease. Virology Journal. 2020; 17: 9. https://doi.org/10.1186/s12985-019-1277-7 PMID: 31973727
- Centro de Vigilância Epidemiológica. Boletim Epidemiologico da febre amarela—21/01/2019. 2019 Dec. Available: http://www.saude.sp.gov.br/resources/cve-centro-de-vigilancia-epidemiologica/areasde-vigilancia/doencas-de-transmissao-por-vetores-e-zoonoses/doc/famarela/2019/fa19\_boletim\_ epid\_210119.pdf
- Centro de Vigilância Epidemiológica. Boletim Epidemiologico da febre amarela—18/11/2019. 2019 Dec. Available: http://www.saude.sp.gov.br/resources/cve-centro-de-vigilancia-epidemiologica/areasde-vigilancia/doencas-de-transmissao-por-vetores-e-zoonoses/doc/famarela/2019/fa19\_boletim\_ epid\_1811.pdf
- 12. Centro de Vigilância Epidemiológica. Boletim Epidemiologico da febre amarela—02/01/2018. 2019 Dec. Available: http://www.saude.sp.gov.br/resources/cve-centro-de-vigilancia-epidemiologica/areasde-vigilancia/doencas-de-transmissao-por-vetores-e-zoonoses/doc/famarela/fa18\_0201boletim\_ epidemiologico.pdf
- Cunha M dos P, Duarte-Neto AN, Pour SZ, Ortiz-Baez AS, Černý J, Pereira BB de S, et al. Origin of the São Paulo Yellow Fever epidemic of 2017–2018 revealed through molecular epidemiological analysis of fatal cases. Sci Rep. 2019; 9: 1–10. https://doi.org/10.1038/s41598-018-37186-2 PMID: 30626917

- QGIS Development Team. QGIS Geographic Information System. Open Source Geospatial Foundation Project. 2018. Available: http://ggis.osgeo.org
- Gubler DJ, Kuno G, Sather GE, Velez M, Oliver A. Mosquito cell cultures and specific monoclonal antibodies in surveillance for dengue viruses. Am J Trop Med Hyg. 1984; 33: 158–165. https://doi.org/10. 4269/ajtmh.1984.33.158 PMID: 6364855
- 16. Drosten C, Göttig S, Schilling S, Asper M, Panning M, Schmitz H, et al. Rapid detection and quantification of RNA of Ebola and Marburg viruses, Lassa virus, Crimean-Congo hemorrhagic fever virus, Rift Valley fever virus, dengue virus, and yellow fever virus by real-time reverse transcription-PCR. J Clin Microbiol. 2002; 40: 2323–2330. https://doi.org/10.1128/jcm.40.7.2323-2330.2002 PMID: 12089242
- Domingo C, Patel P, Yillah J, Weidmann M, Méndez JA, Nakouné ER, et al. Advanced yellow fever virus genome detection in point-of-care facilities and reference laboratories. J Clin Microbiol. 2012; 50: 4054–4060. https://doi.org/10.1128/JCM.01799-12 PMID: 23052311
- Quick J, Grubaugh ND, Pullan ST, Claro IM, Smith AD, Gangavarapu K, et al. Multiplex PCR method for MinION and Illumina sequencing of Zika and other virus genomes directly from clinical samples. Nat Protoc. 2017; 12: 1261–1276. https://doi.org/10.1038/nprot.2017.066 PMID: 28538739
- Quick J, Loman NJ, Duraffour S, Simpson JT, Severi E, Cowley L, et al. Real-time, portable genome sequencing for Ebola surveillance. Nature. 2016; 530: 228–232. <u>https://doi.org/10.1038/nature16996</u> PMID: 26840485
- 20. Li H, Durbin R. Fast and accurate short read alignment with Burrows-Wheeler transform. Bioinformatics. 2009; 25: 1754–1760. https://doi.org/10.1093/bioinformatics/btp324 PMID: 19451168
- Loman NJ, Quick J, Simpson JT. A complete bacterial genome assembled de novo using only nanopore sequencing data. Nat Meth. 2015; 12: 733–735. https://doi.org/10.1038/nmeth.3444 PMID: 26076426
- Katoh K, Standley DM. MAFFT multiple sequence alignment software version 7: improvements in performance and usability. Mol Biol Evol. 2013; 30: 772–780. <u>https://doi.org/10.1093/molbev/mst010</u> PMID: 23329690
- Darriba D, Posada D, Kozlov AM, Stamatakis A, Morel B, Flouri T. ModelTest-NG: A New and Scalable Tool for the Selection of DNA and Protein Evolutionary Models. Mol Biol Evol. 2019. <u>https://doi.org/10.1093/molbev/msz189 PMID</u>: 31432070
- 24. Stamatakis A. RAxML version 8: a tool for phylogenetic analysis and post-analysis of large phylogenetes. Bioinformatics. 2014; 30: 1312–1313. https://doi.org/10.1093/bioinformatics/btu033 PMID: 24451623
- Drummond AJ, Suchard MA, Xie D, Rambaut A. Bayesian phylogenetics with BEAUti and the BEAST 1.7. Mol Biol Evol. 2012; 29: 1969–1973. https://doi.org/10.1093/molbev/mss075 PMID: 22367748
- Ayres DL, Darling A, Zwickl DJ, Beerli P, Holder MT, Lewis PO, et al. BEAGLE: An Application Programming Interface and High-Performance Computing Library for Statistical Phylogenetics. Syst Biol. 2012; 61: 170–173. https://doi.org/10.1093/sysbio/syr100 PMID: 21963610
- 27. Drummond AJ, Ho SYW, Phillips MJ, Rambaut A. Relaxed Phylogenetics and Dating with Confidence. PLOS Biology. 2006; 4: e88. https://doi.org/10.1371/journal.pbio.0040088 PMID: 16683862
- Gill MS, Lemey P, Faria NR, Rambaut A, Shapiro B, Suchard MA. Improving Bayesian population dynamics inference: a coalescent-based model for multiple loci. Mol Biol Evol. 2013; 30: 713–724. https://doi.org/10.1093/molbev/mss265 PMID: 23180580
- Lemey P, Rambaut A, Welch JJ, Suchard MA. Phylogeography Takes a Relaxed Random Walk in Continuous Space and Time. Mol Biol Evol. 2010; 27: 1877–1885. <u>https://doi.org/10.1093/molbev/msq067</u> PMID: 20203288
- Pybus OG, Suchard MA, Lemey P, Bernardin FJ, Rambaut A, Crawford FW, et al. Unifying the spatial epidemiology and molecular evolution of emerging epidemics. PNAS. 2012; 109: 15066–15071. <u>https:// doi.org/10.1073/pnas.1206598109</u> PMID: 22927414
- Dellicour S, Rose R, Faria NR, Lemey P, Pybus OG. SERAPHIM: studying environmental rasters and phylogenetically informed movements. Bioinformatics. 2016; 32: 3204–3206. <u>https://doi.org/10.1093/</u> bioinformatics/btw384 PMID: 27334476
- 32. Delatorre E, de Abreu FVS, Ribeiro IP, Gómez MM, dos Santos AAC, Ferreira-de-Brito A, et al. Distinct YFV Lineages Co-circulated in the Central-Western and Southeastern Brazilian Regions From 2015 to 2018. Front Microbiol. 2019;10. https://doi.org/10.3389/fmicb.2019.00010 PMID: 30728810
- **33.** Casos de febre amarela silvestre em residentes do estado de São Paulo, 2007–2008. BEPA Boletim Epidemiológico Paulista (Online). 2008; 5: 12–15.
- Saad LDC, Barata RB, Saad LDC, Barata RB. Yellow fever outbreaks in São Paulo State, Brazil, 2000– 2010. Epidemiologia e Serviços de Saúde. 2016; 25: 531–540. https://doi.org/10.5123/S1679-49742016000300009 PMID: 27869924

- Nascimento RA do, Schiavetti A, Montaño RAM. An assessment of illegal capuchin monkey trade in Bahia State, Brazil. Neotropical Biology and Conservation. 2013; 8: 79–87–87. https://doi.org/10.4013/ nbc.2013.82.03
- Faria NR, Suchard MA, Abecasis A, Sousa JD, Ndembi N, Camacho RJ, et al. Phylodynamics of the HIV-1 CRF02\_AG clade in Cameroon. Infect Genet Evol. 2012; 12: 453–460. <u>https://doi.org/10.1016/j.meegid.2011.04.028 PMID: 21565285</u>
- Culot L, Pereira LA, Agostini I, de Almeida MAB, Alves RSC, Aximoff I, et al. ATLANTIC-PRIMATES: a dataset of communities and occurrences of primates in the Atlantic Forests of South America. Ecology. 2019; 100: e02525. https://doi.org/10.1002/ecy.2525 PMID: 30317556
- Arruda DM, Fernandes-Filho EI, Solar RRC, Schaefer CEGR. Combining climatic and soil properties better predicts covers of Brazilian biomes. The Science of Nature. 2017. <u>https://doi.org/10.1007/</u> s00114-017-1456-6 PMID: 28324174
- 39. Vainio J, Cutts F. Yellow fever. World Health Organisation, Geneva; 1998.
- 40. WHO | Yellow fever–Brazil. In: WHO [Internet]. [cited 14 Aug 2019]. Available: http://www.who.int/csr/ don/18-april-2019-yellow-fever-brazil/en/
- Dellicour S, Rose R, Faria NR, Vieira LFP, Bourhy H, Gilbert M, et al. Using Viral Gene Sequences to Compare and Explain the Heterogeneous Spatial Dynamics of Virus Epidemics. Mol Biol Evol. 2017; 34: 2563–2571. https://doi.org/10.1093/molbev/msx176 PMID: 28651357
- 42. Instituto Brasileiro de Geografia e Estatística. Municipal Network. 2019. Available: https://www.ibge.gov.br/geociencias/organizacao-do-territorio/15774-malhas.html?=&t=downloads
- Moreira-Soto A, Torres MC, Mendonça MCL de, Mares-Guia MA, Rodrigues CD dos S, Fabri AA, et al. Evidence for multiple sylvatic transmission cycles during the 2016–2017 yellow fever virus outbreak, Brazil. Clinical Microbiology and Infection. 2018; 24: 1019.e1–1019.e4. https://doi.org/10.1016/j.cmi. 2018.01.026 PMID: 29427798